

This is a repository copy of *The interaction of hierarchical and socializing accountability* and the emergence of intelligent accountability in a classroom – a critical analysis.

White Rose Research Online URL for this paper: https://eprints.whiterose.ac.uk/198054/

Version: Accepted Version

Article:

Arikan, O. orcid.org/0000-0002-1916-2013 (2023) The interaction of hierarchical and socializing accountability and the emergence of intelligent accountability in a classroom – a critical analysis. Accounting, Auditing and Accountability Journal, 36 (7/8). pp. 1763-1789. ISSN 0951-3574

https://doi.org/10.1108/AAAJ-01-2022-5646

© 2023, Emerald Publishing Limited. This is an author-produced version of a paper subsequently published in Accounting, Auditing & Accountability Journal.

Reuse

Items deposited in White Rose Research Online are protected by copyright, with all rights reserved unless indicated otherwise. They may be downloaded and/or printed for private study, or other acts as permitted by national copyright laws. The publisher or other rights holders may allow further reproduction and re-use of the full text version. This is indicated by the licence information on the White Rose Research Online record for the item.

Takedown

If you consider content in White Rose Research Online to be in breach of UK law, please notify us by emailing eprints@whiterose.ac.uk including the URL of the record and the reason for the withdrawal request.



ACKNOWLEDGEMENTS: I appreciate the constructive comments of two anonymous reviewers and suggestions by Carolyn Cordery (Guest Editor). I also thank Geraldine Chopra, Ian Cornelius, Sylvia Dempsey, Peta Myers, Stewart Symth, David Yates, and participants at the 2022 BAFA Accounting Education SIG Conference in Glasgow for their helpful comments and suggestions. I am grateful to George Feiger who inspired me to write a diary that later became the foundation of this study. Special thanks go to Emine Zinnur Kilic, who constantly supported me through the difficult times during which I wrote this manuscript, despite her own difficulties.

The interaction of hierarchical and socializing accountability and the emergence of intelligent accountability in a classroom – a critical analysis

Abstract

Purpose – This study investigates the impact on organizational members of team marks and peer feedback in a classroom as an organizational setting, where equals were engaged in a hierarchical form of accountability. It uses Roberts's framework of hierarchical, socializing, and intelligent forms of accountability, and discusses the viability of intelligent accountability in higher education, given the accountability structure for academics.

Design/methodology/approach – Autoethnography based on excerpts from the lecturer's diary.

Findings –The blurred boundaries of hierarchical and socializing forms of accountability create both tensions and kinships for students, and these two forms of accountability constantly impact on each other. Although the accounting tools have an individualizing effect on some students, several examples of intelligent accountability are uncovered. It is concluded that academia's audit culture, which focuses on immediate outcomes, and academics' everincreasing workloads make successful innovations less likely.

Originality/Value — This study contributes to the accountability literature in revealing a constant dynamic between hierarchical and socializing forms of accountability through examination of a unique setting in which the boundaries between the two are completely blurred. By empirically examining how accounting individualizes and how intelligent accountability emerges, this study contributes to the limited empirical literature on the impact of accountability on individuals, and particularly to studies of classrooms as organizations, with implications for education policies.

Keywords Hierarchical, socializing, and intelligent forms of accountability; Boundary maintenance in organizations; Audit culture in Higher Education; Peer feedback in teams; Pandemic; Virtual classroom

1. Introduction

In this study, accountability mechanisms to enhance learning in a classroom and in broader higher education (HE) settings are critically discussed through an autoethnographic account. Its context is a module with team assignments, where peers gave each other scores based on their engagement with the assignment, and these scores formed part of the students' overall module marks. Both the teamwork marks and peer scores were accounting tools to make students accountable for their engagement with these assignments. Amongst various definitions of accountability in the literature, in this setting I define it as being "called to account" (Mulgan, 2000). Thus, my students were being called to account for their efforts on the team assignment, not only by the instructor but also by their teammates. This paper evaluates these accountability tools, and particularly the peer feedback, using Roberts's (1991, 1996, 2009, 2018) theoretical framework of hierarchical, socializing, and intelligent forms of accountability. It also discusses how the hierarchical accountability system for academics in HE's so-called audit culture (Radice, 2013; Strathern, 2003) inhibits intelligent accountability and successful innovation in learning and teaching.

The case took place in the second term of the 2020–2021 academic year, when all teaching was online. The team and feedback structures in my (virtual) classroom meant that the whole classroom and each team qualified as organizations (Cohen, 1976). Several studies have examined classrooms as organizations in HE settings (see, e.g., Cohen, 1976; Mezoff *et al.*,

1979; Pendse, 1984), while many others have critically examined how accountability structures in organizations impact on both account giver and demander (see, e.g., Munro, 1996; Roberts, 1991, 1996, 2009, 2018; Strathern, 2003). However, no previous study appears to have investigated how accountability structures embedded in a module impact on members of the module as an organization. This paper fills that void. It examines the impact of accountability mechanisms in the form of team marks, and particularly peer feedback, on members of the classroom as an organization, namely on students and the instructor.

The study is important for several reasons. First, when student equals give each other feedback that becomes a part of their hierarchical accountability and affects their final grades, the boundaries of hierarchical and socializing forms of accountability become completely blurred, creating both tensions and kinships. This unique setting allowed me to examine the dynamics between these two forms of accountability, and determine how they constantly impacted on each other. Although extant literature accepts that these two forms of accountability may happen simultaneously in the same place, serving similar functions, it does not empirically examine the constant dynamic between the two. Examining accountability in this unique context allowed me to uncover a dimension of accountability—the constant interaction between the two forms—which will serve to enhance accountability in organizations (Sinclair, 1995).

Second, the study contributes to the literature by answering a call for empirical studies of different accountability structures (e.g., Jacobs, 2016) and the mechanisms through which accounting measures operate within these structures (Aleksovska *et al.*, 2019). It contributes particularly to the limited empirical research stream on the individualizing impacts of hierarchical forms of accountability and the emergence of intelligent accountability (see Grisard *et al.*, 2017; O'Regan *et al.*, 2021).

Third, although several studies of team-based learning incorporating peer feedback reflect on the practical issues stemming from that method (e.g., (Lane, 2012; Saldivar, 2015; Stein *et al.*, 2016)), this appears to be the first critical accountability study of a classroom as an organizational setting. Critical accounting literature examines the experiences of individuals in accountability contexts (Lai *et al.*, 2018) and investigates some ethical issues connected with specific accountability forms (Messner, 2009). This paper examines the classroom as an organization through the lens of the accountability literature, and critically analyzes interactions within it, as well as surrounding ethical issues such as avoidance of truthful reporting, favouritism, and shifting the blame to others, using hierarchical, socializing and intelligent forms of accountability as a theoretical framework.

In addition to its theoretical contributions, the paper offers several practical inputs. First, it is about a teaching innovation that appears to have worked well for that module, despite the difficulties in that year. The study therefore reflects on the sorts of policy changes that may need to be considered to enable academics to devise innovations that benefit students. Second, the paper gives an account of how individuals worked in teams when face-to-face interaction was impossible owing to a lockdown during the COVID-19 pandemic, and therefore contributes to our understanding of the social implications of the pandemic.

In this study, I use autoethnography as a methodology, and present my insights as an academic in an HE setting, thereby contributing to the autoethnographic literature on academics. I do so by using excerpts from my diary. This methodology allows me to produce accounts of my daily life and make meanings of these accounts (Ahrens and Mollona, 2007; Allbon, 2012; Holman-Jones, 2005).

3

-

¹ The diary entries are not solely about my teaching, and cover a longer time frame than that referred to in this study. Only relevant excerpts written while the module was being prepared and delivered were examined for this study, and some details have been removed to ensure that the students described in them cannot be identified by others, with very minor changes to enhance clarity.

I find that team marks and peer evaluation were necessary accounting tools to maintain the teams' organizational boundaries. However, both I, as the instructor, and the students initially struggled with this new accountability system. In particular, the blurred boundaries of hierarchical and socializing forms of accountability created tensions for and resistance from some students, and also had some individualizing effects. For example, students felt excluded when their peer feedback average suggested that their contribution had been below average. However, there were also several examples of intelligent accountability.

In the remainder of the paper, Section 2 establishes the setting and Section 3 explains the methodology. Section 4 describes how team and peer evaluation marks worked as complementary accounting tools to maintain the teams' organizational boundaries. Section 5 analyzes the dynamics between hierarchical and socializing forms of accountability. Section 6 gives examples of the individualizing effects of the hierarchical form of accountability, and Section 7 describes how some instances of intelligent accountability emerged in the classroom. Section 8 discusses some policy implications regarding lecturers' accountability structures and their impact on teaching innovations, and Section 9 draws some conclusions.

2. Context

As I was getting ready to teach a postgraduate module in the 2020–2021 academic year, I was feeling bad for my students, knowing that they were likely to be feeling isolated as all modules were going to be online. During that period, England was in a lockdown, with all hospitality and non-essential shopping outlets closed, and indoor gatherings other than religious services and small weddings banned (Brown *et al.*, 2021).

I already had some assignments in mind to improve student learning, and decided to form teams for these assignments so that the students could socialize with each other whilst working on their assignments, to mitigate the alienation brought about by the pandemic. One problem was the possibility of free riding. I decided to tackle it by assigning a proportion of the module marks to feedback from students' peers on their contributions to the assignments. There was some information on my university's website about a cloud application (henceforth referred to as "the application") for peer feedback. The virtual learning tool used by my university also facilitated team collaboration, for example by allowing teams to use webinar rooms, post documents, and exchange emails. I formed the teams and configured the application for the peer feedback.

The application allowed instructors to set up visibility options for the feedback. I chose the option where feedback would be anonymous to the students (they would see the feedback but could not see who had written it), whereas I would be able to see who had given what feedback and to whom. I set up my feedback sessions so that students would provide (1) numerical feedback to each of their peers, (2) narrative feedback explaining how they felt their peers had done on that specific assignment, and (3) narrative feedback explaining how fair they thought their feedback had been from the previous assignment.

The numerical feedback in the application was anchored on equal contribution (E). Students had to state whether their peers had contributed equally to the average, or had contributed less (E-) or more (E+) than the average. If the peer had not contributed at all, students had to note it as a 0 contribution. I instructed the students to follow the template given in Table 1, and explained how the mark they gave (column 1) would correspond numerically with the peer evaluation scores for their peers (column 2).

<INSERT TABLE 1 ABOUT HERE>

Students would get 60% of their assignment mark from the team mark (which would be the same for each member of their team) and the rest would be based on the peer mark.

Therefore, students would get exactly the same mark as their team mark if their contribution to the assignment was equal to the team average (E, corresponding with 100%), a lower mark than their team mark if they had contributed less than the average (E-, contributing to a mark below 100%), or a higher mark if they had contributed more than the average (E+, contributing to a mark above 100). Some students would inevitably enjoy the benefit of working with stronger students and receive a team mark that they would not have obtained by themselves, but their marks would be lower than the team mark if, according to their peers, they had contributed less than the average. Students would receive a 10% penalty on the overall score if they did not provide feedback.

The application normalized the scores to avoid score inflation and deflation in such a way that for each score giver, the sum of contributions was equal to the product of the number of team members and Equal Share (nXE). This meant that if a member gave E+10% to all participants, the application converted these scores to E, as the scores meant that this member believed that each teammate had contributed equally. This was an important mechanism because it made the scoring a zero-sum game: for each student who received an E+, there must be a corresponding student who received an E-. The application gave an average score for each member, called the "perceived contribution," which was the average of the normalized scores given by each peer to the team member. The application allowed instructors to moderate the peer marks where they felt this was appropriate, although I rarely used this facility. I increased some students' scores from 0 to E where the score of 0 had been given by peers because the students had not engaged in an assignment, but they had valid reasons for not doing so, such as bereavement, hospitalization, or sickness. As explained in Section 5, in some instances I also moderated marks when it was clear that the peer mark was not entirely truthful.

To create some competition amongst the teams, I announced that at the end of term, members of the team with the highest average would be allocated a bonus of five points to their individual marks (marks in which the peer scores were embedded). There were 10 teams, each comprising five members. Students had six team assignments during the term, the first of which was a mock assignment. This was marked and was subject to peer feedback like the following assignments, and detailed feedback on their work was given to each team, but the marks for the mock assignment did not count towards the students' final marks. The aim was to show the students how the system worked. The average of the four highest of the five remaining individual marks would count as 10% of students' total marks for the module.

The next section discusses the methodology through which I analyzed interactions in this setting.

3. Methodology

I use autoethnography as my methodology. Autoethnography is a qualitative methodology based on autobiography, and uses the researcher as an object (Boje and Tyler, 2009; Ellis and Bochner, 2000; Haynes, 2011; Humphreys, 2005) to draw attention to the researcher's sense making (Edelman, 1996). This paper uses excerpts from my diary about the module, which I taught between January and March 2021.² Reflecting on my diaries written a while previously, I recall and retell my personal experiences in my classroom setting (Allbon, 2012) through the lens of the critical accountability literature. Team assignment and peer feedback, as accountability tools, exposed both the students' and my own vulnerabilities, and my account as a researcher reflects myself as a vulnerable observer (Behar, 1997; Doloriert and Sambrook, 2009).

² Being autoethnographic research, the study does not include any direct quotations from students. Rather, it includes reflections on students viewed through my own lens, based on my diary entries.

Autoethnography has been criticized as a research methodology for being narcissistic, selfindulgent/confessional, emotional, and unrigorous, as self-accounts are ultimately biased data (Allbon, 2012; Anderson, 2006; Atkinson, 1997; Ellis, 2009; Etherington, 2004; Holt, 2003). However, rather than worrying too much about accuracy, autoethnographers aim to produce accounts that are grounded in daily lives and give meanings to organizational practices (Holman-Jones, 2005). Autoethnographic work necessitates writing about others, as well as writing about oneself (Ellis, 2009). As a personal account, this autoethnographic work reflects on my personal thoughts and feelings, and embraces subjectivity (Ellis and Bochner, 2000), unlike positivistic research which compares propositions against an objective reality (Armstrong, 2008). Although there is a focus on students in this work, the story I tell is my story rather than theirs (Adams, 2008). The reflections are my subjective conclusions on the events, distilled through my experience and the critical accountability lens, which I use to retell events that happened previously (Ellis and Bochner, 2000). Therefore, with this work, although I am inviting readers to trust my narrative, I do not force them to do so, because they are always entitled to challenge it (Yu, 2021). With this personal account, I aim to encourage readers to reflect on their own personal experiences, and thereby establish a sense of connectedness with them (McKie, 2020; Ricoeur, 1995). The influence of my partiality on how I interpret the events inevitably charges my account with emotions, which will provoke readers to think and rethink their own positions, and catalyze change (Holman-Jones, 2005).

Each section of this paper begins with a critical literature review that synthesizes the accountability concepts with which I intend to engage, followed by an observations subsection based on my diary entries that establishes the settings for these concepts in my classroom, and analyzes interactions in these settings through the critical lens of the accountability literature discussed previously. Discussing the critical literature and separately reflecting critically on the literature based on observations is ubiquitous in autoethnographic studies (e.g. Haynes, 2013; Sadler and Wessels, 2019).

The next section explains how team and peer marks were complementary accounting tools necessary to manage the boundaries of the teams in my classroom as organizations.

4. Accounting tools as boundary maintenance mechanisms in organizations

Organizations are groups of people who set goals, make decisions, allocate work, recruit, motivate, control, and reward members (Cohen, 1976). Organization is a continuum: some settings are "organizations" more than others (Pendse, 1984). Following Cohen's (1976) definition of the organization mentioned previously, each team in my classroom was set up as an organization of which I was the supervisor: its members were recruited; the goal was to perform well in the assignments, which were instrumental to the wider goal of learning the module content; and to achieve that goal, the (subordinate) team members allocated work between them. Some accounting tools were used to motivate, control, and reward the team members in my classroom.

Organizations are open systems, as their members are influenced by the outside world, and this influence may interfere with organizational goal congruence (Llewellyn, 1994; Turner, 1990). Accountability mechanisms within an organization are what makes governing possible (Roberts and Scapens, 1985), and accounting tools are designed to discharge accountability to members of the organization (Atkins and Maroun, 2018) in order to discipline them (Agyemang *et al.*, 2019; Rahaman *et al.*, 2010). Introducing such mechanisms into the organization is only possible after drawing the organizational boundaries, to determine who within the organization will be held accountable to it and for what, or in other words define the organizational goals (Power, 2018). An important governance issue for organizations is to avoid shirking, which impacts adversely on fulfilling the organization's aim, as individuals are known sometimes to exert less effort on a task when they are part of a team than when they are

on their own (Kou and Stewart, 2018). This is difficult to detect or avoid in teams without organizational accountability mechanisms (Alchian and Demsetz, 1972).

Shirking in team assignments in classroom settings is a well-reported problem known as "free riding," and peer evaluation is one of the most prominent accountability mechanisms for detecting and controlling shirking in such settings (Hall and Buzwell, 2013; Topping, 1998). These mechanisms serve as binding structures that unify organizations and thus facilitate goal congruence. Accountability mechanisms also help to separate an organization from its environment. For example, financial reporting differentiates one organization from another by quantifying its assets and liabilities, showing it as a distinct financial entity (Llewellyn, 1994).

The next section discusses how two accounting tools were used to manage the boundaries of the teams in my classroom.

4.1. Observations: accounting tools as a boundary maintenance mechanism in organizations

When I set up the teams, I was aware, that each team, like other organizations, was an open system (Llewellyn, 1994), as its members would be influenced by the outside world (Llewellyn, 1994; Turner, 1990). The students' propensities, cultures, backgrounds in the subject being taught, locations, and time zones (many were outside the UK owing to the pandemic), among other factors, would affect the effort they would put into the assignments. Therefore, I did not necessarily expect all team members to do their best for their team to get the best results possible: some would try less, and some might even try to do nothing at all. I had some worry that not everyone would work towards the team's organizational goal of doing well in the assignment, as mentioned in an excerpt from my diary written on December 28:

Teaching prep is taking forever as I am trying this different thing, teamwork, for students not to feel alienated...I am worried about free riding so trying to come up with a system to prevent it.

If free riding were to become ubiquitous, the teams would lose their properties as organizations. Some individual students rather than teams as a whole would be trying to get good marks from the assignments; thus, these students would be the only ones learning the materials, which would prevent achievement of the organizational goal of the whole class learning well. There would be frustrations in teams, which would alienate students from each other and disrupt the teams. Therefore, I had to manage the boundaries of the teams as organizations to ensure their unity and mitigate distortions that might preclude students from working as a team. Indeed, as mentioned above, free riding is well-known to be problematic even with limited number of students who are known to each other (Hall and Buzwell, 2013).

Accounting information, and specifically performance indicators in the form of (1) perceived contribution scores for each team member (see Section 2) and (2) team marks, were my tools for managing these boundaries (Llewellyn, 1994). These were both thresholds differentiating each team as an organization from its environment, and binding structures that unified each team as an organization (Llewellyn, 1994) by clarifying goals and objectives (Marginson and Ogden, 2005). Team members contributed only to their own teams and were awarded the mark of their team only, thus differentiating each team from the others. On the other hand, peer feedback aimed to motivate team members to make their best efforts in their teams, binding each team as an organization.

In addition to making students' efforts in their assignments visible to me (Roberts and Scapens, 1985), peer marks also prompted the students to exert effort in their assignments (Foucault, 1979, 2014, 2018). The vast majority of students consistently contributed well to their team assignments amidst many difficulties arising from the pandemic such as being unable to meet face-to-face, different time zones, bereavement and hospitalization in immediate and close families, and being infected with COVID. In addition, I was very generous

to students who did not contribute to the assignment: even when students showed zero contribution, they would get 60% of their team's mark. One student's case was very interesting. This student did not engage in the module in any other form, but consistently engaged in the teamwork which was worth only 10% of the overall mark. This is an excerpt from my diary entry on March 1:

I am surprised about AAA³...This is a student who never attends my webinars. They⁴ also did not take any of the individuals tests, so have already lost 30% of the mark, but they engage with their team in every assignment and receive satisfactory peer feedback. I wonder if it is the peer pressure coming from the peer evaluations that prompts them to work on these assignments?

This student would probably not have contributed to the assignments had there been no peer feedback that made their efforts more visible to me, to their peers and to themselves. They did not engage in other forms of assessment when this visibility was not present.

As detailed in the next section, pressure from the peer evaluation marks, which was the formal, hierarchical form of accountability, was not the only reason for students to do well. While working together, students recognized their importance to each other. This socializing form of accountability is also likely to have played a role in student AAA's efforts outlined above. The next section discusses these two forms of accountability in my classroom.

5. Hierarchical and socializing forms of accountability

The accountability literature defines hierarchical accountability as a formal accountability structure in which rules are set up, overseen, and enforced by a supervisor (Roberts, 1996). This form of accountability is often contrasted with the socializing form of accountability, which is between equals and informal (Roberts, 1991, 1996, 2018). This latter form arises not from formal rules but from individuals' realization that they make a difference to each other (Roberts, 2018). Roberts (1991) argues that hierarchical forms of accountability originate in the workplace, whereas socializing forms originate from interactions outside the workplace (e.g., work commutes, lunches, etc.). He bases this dichotomy on Habermas's (1971) definitions of work as instrumental action shaped by formal rules, and interaction as communicative action governed by consensual norms.

Roberts (1991) further argues that hierarchical and socializing forms of accountability have entirely different functions. Accounting produces a hierarchical form of accountability only, and in doing so precludes mutual understanding, as knowledge gained from accounting information is imposed on individuals by their superiors to instrumentalize success. In contrast, socializing forms of accountability encapsulate active understanding of each other (Roberts, 1991). This dichotomy is also based on Habermas's (1985) distinction between actions based on success, which Roberts identifies as those that take place under the hierarchical form, and actions based on understanding, which Roberts identifies as those that take place under the socializing form.

Other scholars disagree that these forms of accountability should necessarily take place in the different settings of work versus outside of work. For example, Lindkvist and Llewllyn (2003) argue that individuals may engage in reflective communications in work settings whilst performing their work, while some communications between peers outside work settings may focus on work success (see also Agostino and Arnaboldi, 2018). Some scholars also disagree about the functional dichotomy between hierarchical and socializing forms of accountability.

³To protect student confidentiality, student names in my diary are replaced with triple-letter combinations such as AAA or BBB, each combination reflecting a unique student.

⁴ To further ensure student confidentiality, gender pronouns in my diary are replaced with third-person pronouns.

For example, Lai *et al.* (2014) observe that during the Italian flood of November 2010, accounting tools (seen by scholars as a hierarchical accountability mechanism), such as damage estimations, fostered mutual understanding and dialogue, and created a sense of interdependency between victims and other individuals involved, which are known as functions of the socializing form of accountability in Roberts's framework. Rather than a dichotomy, Lindkvist and Llewllyn (2003) view the two forms of accountability as intertwined processes with continually shifting emphases (see also Cordery *et al.*, 2019).

Scholars agree that each form of accountability has advantages and disadvantages. For example, while Roberts (1991) argues that the hierarchical form has some individualizing effects on members of an organization, which may be balanced by the socializing form, the latter may also have adverse effects, such as favoritism, which may be balanced by the formal, hierarchical form. Similarly, while Lindkvist and Llewllyn (2003) agree that the hierarchical form may be solitary, it may also facilitate collaboration; and while the socializing form may foster mutual understanding, it does not always result in rational consensus.

My case is unique for two reasons. First, as it took place during a pandemic lockdown, it was impossible for students to have the usual "face-to-face communications." which are an essential condition for the socializing form of accountability, a concept based on literature developed prior to the pandemic. Students' interactions typically took place behind screens, with their faces unlikely to be visible to each other because students tend to turn off their cameras in online meetings (Oliveira *et al.*, 2021). Furthermore, the students were spread across geographical locations in different time zones. However, as seen below, none of this prevented the socializing form of accountability from taking place among the students.

Second, the students were held formally accountable for their efforts by their peers/equals through peer feedback. Lateral accountabilities in which team members are held accountable to each other are common in decentralized organizations (Lindkvist and Llewellyn, 2003). For example, in some manufacturing facilities, cellular production units are formed in which work quality is surveyed within the units, where members of the units check the quality of products and may reject them as faulty (Munro and Hatherly, 1993). However, my case differs significantly from such cases, as the hierarchical accountability imposed by the peer feedback had a formal impact on students' attainment. At the same time, as peers, students also felt accountable to each other in the socializing form: they were aware that they should be nice to each other and help each other. As the observations below show, this blended structure of hierarchical and socializing forms of accountability created both tensions and kinships, and formed interesting dynamics. Each form of accountability constantly impacted on the other.

5.1. Observations: hierarchical and socializing forms of accountability

The teams' accountability structure had a hierarchical aspect: it was formal (Roberts, 1996), with rules set up, overseen, and enforced by the lecturer, and peer feedback given by equals had official consequences for student attainment. To achieve the team marks, students had to work together and interact with each other regularly. These social ties brought about a socializing form of accountability (Roberts, 1991, 1996), as students realized that they made a difference to each other, and thus worked hard (Roberts, 2018). I was able to see this in the feedback the students provided to each other, as well as in my correspondence with them. Below is an excerpt from my diary entry on February 23:

I am really proud of my students. They are generally giving such positive feedback to each other, and their pride in working as a team is obvious from their feedback. I especially liked one student's remark to BBB, a student going through a difficult time, who sent their best wishes and asked them not to worry about the assignment. I was even more touched when I got an email from BBB today, who said that

although what happened had hit them really hard, they would try to be part of the team assignment this week as it was not fair on their team to do the work alone.

This case shows how BBB understood that they were an important part of their team and felt accountable to their teammates. Their teammate saw a formal/hierarchical accountability mechanism, the peer feedback form, as an opportunity to show their understanding of the situation. Here we see the peers' mutual understanding, a socializing form of accountability, which seems to have originated while working on the team assignments because the students had no other way of socializing with each other owing to the lockdown.

The blended boundaries of two forms of accountability created some tensions for the students. On one occasion, one team member did not hide their frustration from the rest on learning of the bad mark they had been given for the previous assignment, and another member wrote in their narrative feedback that they should not have behaved in that way, as shown in the excerpt from my diary on February 4:

I love how students give each other some very constructive feedback; they even try to teach each other how to be better people. CCC, while acknowledging the great work DDD has produced in a timely manner, also mentioned how they lost their cool on learning of the bad mark the team got in the last assignment. CCC called DDD to act more calmly in the future, pointing out that as they were going to work together until the end of the term, it was important that anger did not get the better of them.

This was an engagement where the hierarchical form of accountability (the team mark) had an impact on the socializing form of accountability (the student did not fulfil their obligation to their peers to be calm and polite in engaging with them), which in turn impacted on the hierarchical accountability (formal feedback given by the peer). This case thus shows, like the one above, the dynamic relationship between the two forms of accountability.

Furthermore, at least one student did not believe that equals could be involved in a system of hierarchical accountability, as reflected in an excerpt from my February 9 diary entry about the results of the mid-term module feedback:

Results are really good, but one student...seems so upset with the team assignment system (they gave so many details that I understood who wrote that feedback). They say that their grade should be a reflection of their own work, not based on feedback from peers.

This was an interesting remark. The peer feedback *reflected* students' work, but this student seemed to be suggesting that their work performance and the peer feedback were mutually exclusive phenomena. In their view, peers could not reflect on each other's work.

In some cases, some team members resisted engaging with hierarchical accountability that required them to disclose peers who made no effort in the assignment, perhaps because they felt a socializing form of accountability to their peers. Here is an excerpt from my diary entry of March 3:

I am really frustrated with a student who is refusing to tell in their feedback that one of their peers is not engaging with the assignment. There are a few students who did not engage with any of the assignments from the outset, which can be seen by the narrative feedback given by most of their peers. But there are a couple of students who have not been giving them a 0. They are giving them very low marks, but this is not enough; by not giving a zero, they are reducing all the engaging

students' marks.⁵ I must moderate these marks to 0 to be fair to the students. They also resisted giving narrative feedback to non-engaging peers. I have reminded the students many times that they must give narrative as well as numerical feedback. I finally took 10% off their marks as a penalty. One student continues not to engage in any narrative feedback.

This account is of an occasion on which some students refused to disclose that their peers were not engaging, at their own and others' expense; they refused to engage with the demands of hierarchical accountability. They probably felt that giving a 0 would be a betrayal of their relationships with their peers. The socializing form of accountability that these students felt for their peers (e.g., being nice to each other) impacted on their engagement with the hierarchical form of accountability (giving peer feedback), which in turn impacted on their marks, with a 10% penalty aimed at inducing hierarchical accountability.

Socializing forms of accountability, such as being nice to each other, do not necessarily mean that niceness comes from pure benevolence. Amongst many reasons for feeling obliged to be nice, the students might be paying back previous "favors," or they might have been intimidated by the peer, or reputation management tactics discussed in Section 6 might have come into place. In addition to bringing benevolence, socializing may also have adverse effects (e.g. Arbin *et al.*, 2021; Roberts, 1991).

This case illustrates that both forms of accountability are needed. Although the socializing form of accountability is beneficial in reminding individuals that they need each other, it may involve favoritism. This adverse effect may be balanced by formal, hierarchical accountability (Roberts, 1991). The two students who refused to give their non-engaging peers 0, as they should have, favored these peers over other members. The formal, hierarchical accountability mechanism made this favoritism visible through the feedback of other peers and allowed me to intervene to ensure fairness.

Finally, one student who complained about free riding turned actually out to have facilitated it. Returning to my journal entry of February 9 about the student who expressed frustration about the team assignments, here is another excerpt from the same entry:

They complain that a few of the group are left doing all the work with all the pressure and the rest of the group gets a free grade. Strangely though, when I checked their previous feedback, I saw that they never gave a bad mark to any of the peers in any of the assignments; the lowest mark they gave to their peers was an E, and the largest range between their lowest and highest mark was 10%.

The whole point of the peer feedback system was to prevent free riding, and this student seemed to be ignoring the connection between the two. It seems that this student was reluctant to give these peers low marks because they felt a socializing form of accountability towards them.

Overall, some students sometimes struggled with the system, which was based on two accounting measures: team marks and peer feedback. Consistent with Roberts (1991, 2018), accounting measures had an individualizing effect on them at times, as discussed in the next section.

6. Individualizing effects of accounting information

Roberts (1991, 2018) argues that accounting information serves hierarchical accountability, allowing outputs to be compared with others and with benchmarks (see also Robson, 1992; Vollmer, 2007). Such comparison has an individualizing effect, as it prompts individuals to

11

⁵ The application automatically allocated the marks of non-engaging students to other team members. When these students were given any mark other than 0, the part that would have been allocated to engaging students decreased.

⁶ Recall that owing to the details the student put in their feedback, I knew who had written it.

compare themselves with others and perceive others as their competitors (Roberts, 1991). Roberts (1991) further argues that being held accountable for one's actions sharpens people's sense of self and desire for recognition, and that when they fail to meet the benchmark, they suffer because their desire for recognition is not met, and their self-worth deteriorates. Hoskin (1996) argues that when students are graded, they may perceive their marks not simply as rating their performance in a particular assessment, but their general ability to do well.

Preoccupation with oneself does not occur only in failures; success brings recognition and acceptance, but both are fragile and can be taken away at any point when expectations are not met (Roberts, 1991). In this individualizing process of accountability, individuals focus on avoiding blame and receiving praise (Roberts, 2018), and may engage in reputation management tactics without necessarily trying to increase their efforts (Roberts, 2003, 2009). All of this inhibits learning (Roberts, 2009).

This discussion reveals that accounting information has an alienating effect on individuals, whether successful or not. The observations below show how the accounting tools used in my classroom made the experience of engaging in teamwork a lonely one.

6.1. Observations: individualizing effects of accounting information

Roberts (2018) argues that hierarchical accountability assesses whether individuals meet advertised criteria, not their uniqueness. In my setting, students were assessed against the criterion of equal contribution, which corresponded with an evaluation mark of 100%. To receive the full team mark or above, students had to show a contribution at least equal to the average of the team.

After the *mock* assignment, several students' remarks revealed that they would be very upset if, contrary to their own belief, their perceived contribution score was below 100. An excerpt from my diary entry of January 20 summarizes the issue:

Students received their mock assignment peer evaluation scores, and several of them were angry about getting a perceived contribution score below 100. They claimed that they had contributed equally, but all of those who sent me a complaint email received scores very close to 100: 99 or 98. I tried to explain this to them. I also tried to explain that these are averages of all the scores given by peers, and they must not be taken personally. For example, if one member contributed significantly more than the others, this would decrease the mark for some others to below average. Some of these students were still adamant that their scores were unfair. I suggested they should discuss this with their teammates.

For students expecting a 100, receiving a 99 or a 98 should not be of much concern, especially when this score was the average of four team members' scores. If anything, they should have been pleasantly surprised that their teams' assessments of them were so accurate. However, this was not the case. Students were upset, even though they knew that this mock assignment had no impact on their official marks. Consistent with Roberts (1991), when students did not meet the benchmark of equal contribution, they suffered from the experience, which alienated them from their team members as they felt resentment towards them. Sometimes their resentment was also channeled to the lecturer, as shown in the cases below.

The diary entry discussed in Section 5 described a student's complaints. Another excerpt from the same diary entry dated February 9 is as follows:

They even say the whole thing seems to be a secondary school-level activity, not masters, and that they thoroughly dislike the module! They are so full of anger.

I decided to have a virtual meeting with this student and sent them a calendar invitation. I explained that thanks to the details they had provided in the feedback, I had realized that the

feedback had been written by them. After our meeting, I wrote the following journal entry, dated February 15:

At one point the student started crying because they were feeling very pressurized about the group work and were frustrated with some of the team members sending their work at the last minute!

As previously mentioned, the total of these assignments contributed 10% towards students' marks for the module, which meant that the contribution to their overall degree classifications was only 0.8%. With five students in each group, assuming equal contribution, each student's non-contribution would have an impact of less than 0.2% on any group member's attainment on the program. In fact, the peers about whom the student complained *were* contributing, but were sending their work late. All in all, what the student was complaining about had a negligible impact on students' overall success, yet the student reacted very emotionally (crying) to the last-minute submissions of some peers. This did not seem rational to me. It suggested that the student was taking the accountability measures established to promote learning in the classroom personally, possibly equating their team's success with their own success, or even their own self-worth (Foucault, 1979; Roberts, 1996). The fact that they did not enjoy the module and the anger they felt suggest that the accountability tools had an individualizing impact on the student.

There were other instances of angry resistance from my students. As explained in Section 2, members of the most successful team would get a five-point bonus, and the top three teams were announced after each team assignment. In one instance, some teams that had previously been consistently in the top three did not come close to satisfying the session's learning objectives and received low marks for the assignment. My feedback to the teams reflected why they had lost marks. The excerpt from my diary entry of February 24 explains what happened next:

Two students [from different teams] who were not happy with their results contacted me and they seem to believe that I am not rewarding critical thinking. One talks about their successes so far, points out that they have a wider understanding of this subject, and say they are worried that I would unfairly give them a lower mark in the final exam than a student who just recites the slides. The other says that a postgraduate degree should aim to bolster critical thinking, and complains that I am discouraging free thinking because I cut marks for citing some external references.

These teams lost marks because their answers did not reflect any critical thinking which, as one of the students rightly suggested, was the main aim of that module. The students seemed to equate citing external sources with critical thinking. The students' reactions suggested that they took the negative feedback on their team assignments personally. Since they knew that the lowest marks would be discounted, they did not have to worry about that assignment because they had received good marks for the previous assignments.

Both students were team leaders, as their consistently high peer scores showed.⁷ In relation to the fragility of recognition and acceptance brought by success (Roberts, 1991), the hierarchical accountability of the peer feedback system made recognition more visible to these students. They consistently received very high peer feedback and team scores, but suddenly their low team marks possibly suggested to them that they had not led their groups in the right direction for that assignment. Roberts (2009) argues that hierarchical accountability correlates with our narcissistic tendency to shift blame, and one strategy to avoid blame is to find a

13

⁷ The teams were not required to select their leaders, but in some teams, leaders naturally emerged as the term progressed.

scapegoat (Hood, 2007). For these students, I was the scapegoat. They blamed me for not rewarding critical thinking, and fully assumed that their approach to the question asked in the assignment was the right one. They did not ask what they could have done differently to do better. Thus, both students engaged in reputation management tactics without necessarily trying to increase their efforts (Roberts, 2003, 2009). Consistent with Roberts (2009), their preoccupation with themselves seems to have inhibited their learning.

Importantly, both students only talked about themselves: how they individually had approached the question, and why their individual answers should be accepted as correct. In other words, they seemed to have forgotten that they were part of a team. Their approaches suggest that the accounting measure of team marks alienated them. They took leader roles in their teams, and judging by the feedback, both teams were functional and were working in good faith. However, in their own minds, these two students seemed to be all alone. Their teams' successes and failures were their own successes and failures. Thus, as an accountability tool, the team mark had an individualizing effect on these students.

The online learning environment and students' inability to be together in the same physical location owing to the lockdown may have exacerbated this alienation. I was just a "voice" to the students behind my screen, as my internet connection was not usually strong enough to turn on my camera. This was probably also the case for students whilst working on their teamwork assignments, or they may even have only been interacting through text messages. Things might have been easier if they had been able to interact face-to-face. However, the accounting tools undoubtedly had some individualizing impact on some students, and the virtual learning environment was not entirely to blame.

Luckily, these accounting tools did not have solely negative impacts on the classroom. There were some good examples of intelligent accountability, as elaborated in the next section.

7. Intelligent accountability

O'Neill (2002), who first coined the term "intelligent accountability," agrees with Roberts (1991) that focusing blindly on accounting numbers has undesirable effects and is of no value. Indeed, research finds that such blind focus leads to poor decisions (Dalla Via *et al.*, 2019). O'Neill (2002) argues that rather than blindly accepting the accounting numbers, we need to actively inquire into what happened to produce these numbers, for example by asking questions and listening to each other. O'Neill (2002) further posits that intelligent accountability is a precursor to well-placed trust, a notion she describes as crucial to our wellbeing, and even to our existence.

Building on O'Neill's (2002) arguments, Roberts (2009) defines intelligent accountability as understanding the relevance or accuracy of indicators of accountability in context through active inquiry. He contrasts this form of accountability with focusing merely on a predetermined set of performance indicators. Roberts (2009) also suggests that active inquiry will improve our understanding of the complex interdependencies beneath accountability indicators, which is vital for understanding how our roles in the organization fit together with accountability measures (Agyemang et al., 2019; Fry, 1995). He elaborates on how intelligent accountability can be achieved by drawing on Butler's (2005) argument that because, as individuals, we are opaque even to ourselves, being fully self-accountable is impossible. Roberts (2009) argues that the first step towards intelligent accountability is understanding our own imperfections, such as this opacity. However, he explains that this cannot be achieved by focusing solely on accounting numbers, as these numbers assume full transparency of self, which is never achievable. He refers to the contradiction that norms not written by them define how individuals should give an account of themselves to others, when even giving account of oneself to the self is impossible (Butler, 2005). In Roberts's (2009) framework, these norms correspond with the hierarchical forms of accountability arising from accounting tools such as performance indicators. Roberts (2009) further argues that focusing solely on accounting numbers will lead to prioritization of defending one's self-image over communication and learning. Only by recognizing our own imperfections, such as our opacity to ourselves and others, can we break this impasse. This recognition creates a narcissistic yet productive wound.

Although both O'Neill (2002) and Roberts (2009) agree that intelligent accountability typically requires face-to-face encounters, they argue that face-to-face relationships alone are insufficient to foster trust in organizations (see also Roberts, 1991). They agree that we still need the hierarchical form of accountability, not only to detect cheating or local collusion, but also, used intelligently, to understand the interdependencies inherent in it, and to learn from it about ourselves and others to advance collectively.

According to Roberts (2009), recognizing our imperfections paves the way to intelligent accountability which can be synthesized in the following steps:

- 1. Recognizing our own imperfections, such as being opaque to ourselves: this allows self-reflexivity by releasing us from pressure to be or appear to be perfect. This, in turn, releases our preoccupation with defending our self-image.
- 2. Accepting the imperfections of others: once we accept that we are imperfect, it is easier for us to accept the imperfections of others. This creates mutual understanding of each other, and appreciation that we are responsible for each other.
- 3. Approaching others as a resource to develop our capabilities, rather than as a threat to ourselves, by asking questions and listening: as the first two steps release us from defending our self-image and help us to appreciate that we need each other to improve, we can stop seeing others as a threat to ourselves. To let others improve ourselves, we need to ask questions and listen. Thus, dialogue is the cornerstone of intelligent accountability.
- 4. Appreciating the importance of accountability mechanisms: once we realize that we are opaque even to ourselves, we understand that we are ignorant about the impact of our behavior on others, and therefore need a system of accountability to be aware of that impact.
- 5. Using accounting tools to advance collectively: as the third and fourth steps allow us to appreciate the importance of accountability mechanisms and approaching others as a resource to develop ourselves by engaging in dialogue with each other, we can incorporate information from accounting tools with information we gather from others. This, in turn, helps us to advance collectively.

The following observations reflect on instances where accounting tools in my classroom fostered some intelligent accountability amongst team members.

7.1. Observations: intelligent accountability

I was happy to see evidence of some intelligent accountability developing in the teamwork, for which the accounting tool of peer feedback was instrumental. This is reflected in an excerpt from my diary entry of February 9 on the mid-term module evaluation results:

I was particularly happy about one remark...One student mentioned how the peer feedback motivated them to work harder and engage more with their teammates in the group work. They liked that the more they engaged with other students to discuss their answers, the more they discovered their own shortcomings, and that broadened their thinking about problems, rather than being confined to their own ideas.

This remark reveals many of the properties of intelligent accountability discussed above, including listening and talking (engagement), understanding one's own imperfections yet not being preoccupied by them, realizing the importance of accounting tools (peer feedback), trusting others, acknowledging that one needs others for advancement, and working together towards that advancement (O'Neill, 2002; Roberts, 2009).

Other instances of intelligent accountability also took place in my classroom. For example, many cases of narrative feedback provided detailed justifications of why a specific mark had been given to a student, such as what the student had worked on, their timeliness, their responsiveness, their communicativeness, and whether they had asked for help in a timely manner when they struggled. Therefore, students' feedback to each other made it easier for them to interact intelligently with the accounting information, by helping each other to understand the complex interdependencies behind it (numerical feedback) rather than being preoccupied with their scores. Thus, it was perhaps unsurprising that throughout the term, the vast majority regularly confirmed that they were happy with their feedback scores for the previous assignment. When they were not, they would write down why they thought the feedback was unfair. They talked about the narrative feedback they had received, contrasted it with what they had done, and explained why they thought the feedback was unfair. In other words, there was dialogue between the students through this hierarchical form of accountability, and dialogue is the cornerstone of intelligent accountability. This worked particularly well in the virtual learning environment, as students had few opportunities for faceto-face discussion. However, similar accounting tools may be valuable in allowing individuals to express themselves and engage in dialogue even when face-to-face encounters are possible.

Students were not the only ones who benefited from intelligent accountability. By interacting with my students, as I did following their complaints about the team assignments, I also engaged in intelligent accountability. Recall that a student complained in their mid-term module feedback about the team assignment, as reflected in the excerpts from my diary entry of February 9 given in Sections 5 and 6. Section 6 also explains that I arranged a meeting with this student to discuss the issue. Another excerpt from my notes of that meeting gives more detail:

The student accepted my meeting invitation one minute past the meeting time and we managed to have a talk...At one point the student started crying... These are such young, naïve individuals!

The student's strong emotional reaction to some team members sending their work at the last minute, which had a negligible impact on their overall attainment, suggested that they were equating the team mark with their own self value, as elaborated in Section 6 (see Foucault, 1979; Roberts, 1996). Seeing this, whilst acknowledging the difficulties of the teamwork, I encouraged the student to see team assignments as a learning experience, rather than letting some poorly performing teammates get them down. I felt that our meeting ended on a positive note with mutual understanding. We exchanged some pleasant emails after the meeting, and they never complained again.

By accepting the invitation, albeit late, the student took a step towards engaging in dialogue, the cornerstone of intelligent accountability. Although the meeting was virtual, it was as close to a face-to-face meeting as possible during the lockdown, with cameras and microphones switched on on both sides. This may have facilitated the dialogue. It was the first time I had seen the student, and may have been the first time the student had seen me, as my internet connection was often not good enough to turn on my camera during the webinars. There were personal touches in the conversation, such as learning that the student was going through a personally difficult time and feeling bad for them about this. However, the moment the student cried was when everything changed for me. I realized at that moment that for the student, this was somehow much more than a team assignment with a negligible impact on their grade. They were taking it much more seriously, and it was hurting them. I then realized what I should do, which I had not done initially when I had first realized that the student was unhappy: remind the student of the essence of these assignments, as nothing more or less than a learning experience. I was also very careful to be sensitive to the student when

communicating this message, by acknowledging the difficulties of team assignments and showing my empathy with their feelings. I also gently pointed out that the peer feedback scores were instrumental in avoiding the free riding about which they complained before, and that in the past they had been a bit too generous to their teammates in their feedback. In doing so, I tried to explain the complex interdependencies beneath the accounting indicators, another aspect of intelligent accountability (Roberts, 2009).

During that conversation I realized two of my previous mistakes. The first was that on initially reading the feedback, I had felt that the student was overreacting and being unfair to me. I had actually lost sleep after receiving that feedback. However, after meeting the student and listening to them, I realized that this was not true. They did not mean to be unfair to me; they had overreacted because they felt under pressure, and did not yet have the diplomatic skills gained from experience. My second mistake was that on reading their feedback, I had totally missed the importance of the assignments to the student (for elaboration on the individualizing effect of these assignments on the student, see Section 6), which precluded the student from seeing the big picture. By realizing this mistake (my own imperfection), I was able to step in and communicate the big picture to the student.

This was not the only meeting that resolved tensions. I met with the two students who claimed that the team marks they had received were unfair (see Section 6), and both meetings ended on a positive note: the students understood why they had lost marks, and I understood that I should have given those teams more detail in my original feedback. I also engaged with many students through emails and messages on the peer feedback system, and encouraged them to talk to their teammates in the few instances where they expressed concerns about their feedback scores. All these efforts to build dialogue seem to have paid off, as complaints about peer feedback dropped significantly by the end of term, and the end-of-term module evaluation indicated a high level of satisfaction with the module.

The above observations suggest that the interaction of hierarchical and socializing forms of accountability fosters intelligent accountability. We need hierarchical accountability since it helps us to understand our strengths and failures. We also need some dialogue to reflect on the feedback from hierarchical accountability, and socializing accountability promotes that dialogue. The new system thus seems to have worked well for my students in the end. The next section discusses what academics need to enable them to bring successful innovations to their classrooms.

8. Reflections for academia

The feedback system described in this study is an example of an educational innovation. Although peer feedback has been widely used in HE, this was its first use on the module I was teaching. It had many innovative qualities, such as its zero-sum nature to prevent gaming the system, penalties for failing to give feedback, the cloud application to facilitate interactions with the system, the competition, and reporting features that brought transparency to everyone involved whilst protecting anonymity. My students' struggles, at least early on, suggested that they did not have prior experience of such a system, so it was a new tool for everybody involved.

I introduced the team assessments into my module for two reasons: to improve learning (the assessment part) and to mitigate the effects of loneliness arising from the pandemic and the accompanying lockdown (the team part). As elaborated in Section 4, the peer feedback scores were integrated into the system, so all students engaged with the experience, rather than a selection who were already self-motivated to learn.

I am in no position to propose that this is the best system, as many alternatives to assessments are available to improve student learning, and I observed some individualizing aspects of the accounting mechanisms. However, the system seemed to work in my classroom

on this occasion. Student satisfaction was high, reflected both in the end-of-term module evaluation and in numerous emails I received from students about how they had enjoyed and benefited from the module. Their attainment was not only very satisfactory, but was also significantly higher than in previous years when these assessments had not taken place. The module evaluation form had a question about whether the pandemic had had a negative impact on students' engagement with the module. The answer was overwhelmingly no, and students reflected that how I had structured the module had impacted positively on their learning during the pandemic. Most importantly, I saw numerous examples of intelligent accountability in my classroom, arising from these assessments and how we engaged with each other in relation to them. The aims of cooperative learning seem to have been achieved: students shared resources (almost all of them contributed their knowledge and effort to the assessments), they provided feedback to each other, and judging by that feedback and the module evaluations, they gained insights into problems by challenging one another (Wilkin, 2022).

Assignment marks and peer scores were measures through which I held students accountable to me and to each other. I tried to achieve this by making students' efforts in team assignments transparent. However, as Strathern (2003) argues, transparency often hides many practices, and a perfect representation of the truth is impossible in any accountability system. O'Neill (2002) suggests that accounting information should be used to limit deception, and that intelligent accountability is only possible through good governance.

In trying to achieve good governance, one of my tools was to pay attention to the accounting information. I read all numerical and narrative feedback to get a sense of whether students were (1) engaging in the teamwork, (2) giving each other constructive and fair feedback, and (3) happy with the feedback they received. In all cases the majority were, although there were inevitably some exceptions, and I may not have seen everything. I stepped in when I found evidence of deception, as explained in Section 5. I also stepped in and started dialogues with students when they were unhappy with peer feedback, where I tried to ensure that they understood how the system worked and encouraged them to engage in dialogue with their teammates (see Section 6). As explained in Section 7, I tried to calm students down when the system proved too stressful for them. I also tried to engage in good governance when marking the team assignments. I prepared a detailed "suggested" solution (emphasizing that students could approach the problems from different angles), and gave detailed and timely descriptive feedback to each team after each assignment. In addition, I diligently wrote my observations in my diary. These entries ultimately constitute data from a classroom team setting to inform team settings in other contexts, and hence contribute to research on accountability.

All these activities were on top of my other responsibilities to my students and to the university, and were therefore difficult to achieve. It meant working more than 12 hours on a typical day. I was able to do all of this, perhaps ironically, because I am on a teaching and research contract, and therefore have a significantly lower teaching load than my colleagues who are on teaching-only contracts; yet the latter are those on whom we depend for most of the teaching delivered in our universities. They are far more experienced in teaching than faculty on research and teaching contracts, and are therefore likely to have better insights than others into the sorts of innovations that students might need; but with such high teaching loads, how can we expect them to be innovative? If we do not devote sufficient time, especially on the first occasion until the system is well established, innovations are bound to be unsuccessful.

However, university lecturers' workloads have been increasing, making such innovations less likely. As detailed below, these increased workloads are largely the result of ever-increasing hierarchical forms of accountability demanded of academics. Increased dependence on this form of accountability, with no dialogue about how educational aims can be better achieved, hinders intelligent forms of accountability and collective advancement of HE, in the UK and around the world (Griffith and Altinay, 2020; Pitman, 2016; Smith and Fleisher, 2011).

Lecturers in the UK went on strike in 2019 (McKie, 2020; BBC, 2021), and another strike decision was made in November 2022 (UCU, 2022a) to protest against, amongst other things, adverse workload conditions (UCU, 2022a). A 2021 survey of members of the UK's University and College Union (UCU, 2022b) indeed demonstrates that staff in HE are working the equivalent of more than two days unpaid per week.

The transformation of HE since the 1990s when government prescription and control increased, referred to as an "audit culture" by academics (Radice, 2013; Strathern, 2003; for implications, see also Amit, 2003; Fillitz, 2003; Gefou-Madianou, 2003; Rimoldi, 2003; Shore and Wright, 2003), introduced audit procedures for academics, such as the Research Excellence Framework (REF) and the Teaching Excellence Framework (TEF).

The REF (formerly known as the Research Assessment Exercise, first conducted in 1986; (Webster, 2012) is used to allocate most institutional funding for research to universities in the UK (Arnold *et al.*, 2018). Its criteria focus largely on publications in high-status journals, which influences academics' careers, as a result of which the importance of activities other than research, including teaching, has decreased for the career advancement of academics on research and teaching contracts (Arnold *et al.*, 2018). This changed somewhat with the introduction of the TEF, which is a government-backed assessment of undergraduate teaching in HE institutions, covering all universities in England and some in Scotland and Wales. Its first results were published in 2017 (Times Higher Education, 2017), and its aim is to rebalance the overemphasis on research brought by the REF, improve the quality of teaching, and provide a basis for students to choose between HE providers (Deem and Baird, 2020). An independent review of TEF criticizes it for not using any direct measure of the quality of teaching, and instead using feedback from final-year students in the National Student Survey (NSS) as its only proxy for teaching quality (Pierce, 2021). TEF is also highly criticized by academics in similar respects (see, e.g., Barkas *et al.*, 2019; Lillejord, 2020).

The NSS, which has run annually since 2005, has been highly controversial in and of itself (Office for Students, 2020). The survey collects data on students' opinions about their courses (e.g., teaching, assessments, and feedback) to inform prospective students' decision making, to help HE institutions to improve their teaching, and to increase public accountability (Office for Students, n.d.). Nevertheless, since its inception, academics have called for it to end, and students have boycotted it. In 2017, the National Union of Students (NUS) instigated a joint boycott with the UCU (Office for Students, 2020), one reason for which was the NSS's use by universities to justify firing staff, cutting courses, and closing departments (UCU, 2016). The NSS has also been criticized for pitching students as consumers, and putting undue pressure on academics by subjecting them to unwarranted surveillance (Office for Students, 2020).

The student-as-consumer model, which was introduced after 1992 (UCU, 2010), invites students to view HE as a market, and in turn emphasizes student satisfaction in course design and delivery, which is criticised for de-professionalising the role of academics, since students, unlike the academics who are educating them, are not experts either in their fields of study, or in the area of pedagogy (Streeting and Wise, 2009). A recent briefing by the Office for Students, the independent regulator of HE in England, includes concern about the credibility of degrees in the UK, suggesting that the student-as-consumer model has caused grade inflation that does not reflect improvements to student achievement (Office for Students, 2022). In this briefing, the Office for Students affirms the existence of grade inflation in the UK over the last decade, and concludes that it is damaging the reputation of English HE.

This increasing focus on positive student feedback and attainment influences us as academics, as we work in an audit environment and our teaching is judged by these short-term

19

⁸ A similar survey, the Postgraduate Taught Experience Survey, is run annually for postgraduate students (Advance HE, 2020.).

performance indicators. The student-as-consumer model seems to have been integrated into academic career progression in the UK. Most, if not all, universities include positive module feedback as one success criterion for academic promotion, and some also include successful module outcomes, such as the number of firsts and upper seconds awarded, and reductions in module failure rates (see, e.g., Swansea University, 2022). The latter is particularly worrying, as academics typically write and mark their own exam papers, and these types of success criteria may exacerbate grade inflation and serve to manage appearances (Cooper and Lapsley, 2021), rather than increasing accountability for delivering high-quality education.

The hierarchical accountability systems described above mainly use assessment scores, as well as student satisfaction scores which are inevitably influenced by their assessment scores. The primary aim of assessments is to judge what is learned, to hold students to account for their learning, and, if necessary, to take actions to improve learning. The secondary use of assessments to hold educators to account for their teaching may undermine the educational aims by introducing incentives such as inflating grades (Office for Students, 2022; O'Neill, 2013). Moreover, increased reliance on these hierarchical accountability mechanisms discourages innovation, as educators fear that non-compliance with the demands of these mechanisms will be punished (Shore and Wright, 2003). For instance, I cannot imagine any university saying, "it is OK that this term's student satisfaction and attainment were low; you came up with a very valuable innovation that will benefit our students in the long run."

In addition to increased reliance on hierarchical accountability mechanisms, the scope for intelligent accountability, which helps make sense of the numbers in these accounting tools (O'Neill, 2002; Roberts, 2009), is decreasing. The UCU's 2021 survey of members (UCU, 2022b) demonstrates that academics spend significantly more time on teaching delivery than five years ago, as well as on audit procedures such as the REF and TEF, and that they spend significantly less time on student consultations, teaching preparation, course review and development, and their own professional development. In other words, whilst time for teaching and audit procedures is expanding, all other activities that would help academics reflect on their performance and improve teaching quality are decreasing.

As discussed in Section 7, understanding the complex interdependencies beneath the accounting indicators through active inquiry and discussion rather than a blind focus on the numbers is important in every organization, but especially so in educational organizations. This is because accounting numbers inevitably simplify the areas evaluated to ease measurement (Power, 2018), but areas that are important for education are difficult to measure or compare. For example, different subjects require different assessments, and their results are not necessarily comparable (O'Neill, 2013). Thus, educational objectives cannot be achieved without proper communication and reflection. However, the current state of the HE system does not allow academics sufficient time to do this, nor do the accountability mechanisms incentivize them to innovate based on their reflections.

By focusing on longer-term outcomes and introducing a system allowing us to give our accounts of what we have done for our students in an intelligent way, policymakers might reflect their trust in us more, as no accountability system can work without trust (O'Neill, 2013).

9. Conclusion

In this study, the accountability mechanisms in a classroom and in broader HE settings to enhance learning are critically discussed. This is an autoethnographic case study about a module in which peers gave each other numerical and narrative feedback based on their engagement with team assignments, and the numerical feedback formed part of the students' marks for the module.

The study concludes that in a classroom setting with peer assignment and feedback, the hierarchical and socializing accountability constantly interact and this interaction fosters intelligent accountability, a form of accountability which is particularly important in educational settings. The results also show that, despite tensions created by the clash of socializing and hierarchical forms of accountability and the individualizing impacts of the accounting information, the teaching method worked well and gave rise to several examples of intelligent accountability in my classroom. However, this result was only achieved through very time-consuming commitment to governance, and that amount of time is typically not available to many academics who do the bulk of HE teaching. Academics' ever-increasing workloads and an accountability structure based on short-term outcomes, such as immediate student evaluations and marks, decrease the likelihood of teaching innovations that would benefit students in the long run. Hence, the paper provides insights for HE policymakers into how to enable such innovations. Particularly, the hierarchical accountability system for academics in HE's so-called audit culture (Radice, 2013; Strathern, 2003) inhibits intelligent accountability and successful innovation in learning and teaching.

By critically analyzing the accountability mechanisms in a classroom as an organizational setting, this study contributes to the literature on classrooms as an organization (e.g., Cohen, 1976; Mezoff *et al.*, 1979; Pendse, 1984) and to the literature on team-based assessments with peer feedback (e.g. Lane, 2012; Saldivar, 2015; Stein *et al.*, 2016). These typically examine how classrooms should be organized to enhance learning and how peer feedback is beneficial to student learning, but do not critically discuss the accountability mechanisms embedded in these settings.

This study answers calls to uncover different dimensions of accountability (Sinclair, 1995) and to empirically examine different accountability structures and mechanisms (Aleksovska et al., 2019; Jacobs, 2016). In this case, peer feedback created a unique setting in which student marks, as a hierarchical accountability tool, were influenced by marks given by equals, who felt some responsibility to their peers arising from the socializing form of accountability, which clashed with their responsibility arising from the hierarchical form of accountability. This setting allowed me to clearly observe the dynamics between these two forms of accountability, and how they constantly impacted on each other. This is an important finding. Although previous literature accepts that these two forms of accountability may interact, occur in the same places, and serve similar functions, it examines them as two separate entities. By showing how intelligent accountability emerged from my setting, despite the individualizing impact of some accountability measures used in it, the study also contributes to the few empirical studies that examine the individualizing impacts of the hierarchical form of accountability and the emergence of intelligent accountability. In other words, by critically evaluating the data from a classroom team setting, this study informs team settings in other contexts. Finally, by describing a teaching innovation that worked well during a COVID-19 pandemic lockdown when face-to-face interaction was impossible, this paper offers some practical inputs and contributes to our understanding of the social implications of the pandemic.

However, this study covers one specific case, so its results cannot necessarily be generalized to other contexts or individuals. Therefore, it is important for future researchers to continue to examine these dynamics in different settings with different complexities (Scobie *et al.*, 2020) and cultural properties that may impact on the success or failure of accountability practices (Leoni *et al.*, 2021). This will uncover more dimensions in which accountability is experienced, which will in turn serve to enhance accountability in organizations (Sinclair, 1995). Future researchers might, for example, examine how to increase instances of intelligent accountability (see Section 7) by using this dynamic, as the dialogue embedded in the socializing form of accountability may be instrumental for using hierarchical accountability in an intelligent way. Conversely, future researchers might also investigate how adverse impacts

of the socializing form of accountability on the formal, hierarchical form of accountability (see Section 5) might be mitigated in organizations. Classrooms are not the only context in which this dynamic occurs (see Section 5), giving rise to other opportunities for researchers to conduct such examinations.

This study is subject to similar caveats to other autoethnographic work, as it is subjective (Ellis and Bochner, 2000). Although the focus is on students in this work, the story I tell is my own rather than theirs (Adams, 2008). Thus, the students' accounts are told from the instructor's perspective. Although I invite readers to trust my narrative, I do not force them to do so because they always have discretion to challenge it (Yu, 2021). Future researchers might extend our understandings of the dynamic between hierarchical and socializing accountability by examining the issue from students' point of view through interviews and focus group discussions. Such methodologies would bring more theoretical insights, and would help academics enhance team-based learning experiences in their classrooms.

References

- Adams, T.E. (2008), "A review of narrative ethics", *Qualitative Inquiry*, Vol. 14 No. 2, pp. 175–194. https://doi.org/10.1177/1077800407304417
- Advance HE (n.d.), "Postgraduate taught experience survey (PTES)", *Advance HE* [website]. https://www.advance-he.ac.uk/reports-publications-and-resources/postgraduate-taught-experience-survey-ptes
- Agostino, D. and Arnaboldi, M. (2018), "Performance measurement systems in public service networks: the what, who, and how of control", *Financial Accountability & Management*, Vol. 34 No. 2, pp. 103–116. https://doi.org/10.1111/faam.12147
- Agyemang, G., O'Dwyer, B. and Unerman, J. (2019), "NGO accountability: retrospective and prospective academic contributions", *Accounting, Auditing & Accountability Journal*, Vol. 32 No. 8, pp. 2353–2366. https://doi.org/10.1108/AAAJ-06-2018-3507
- Ahrens, T. and Mollona, M. (2007), "Organisational control as cultural practice: a shop floor ethnography of a Sheffield steel mill", *Accounting, Organizations and Society*, Vol. 32 Nos. 4–5, pp. 305–331. https://doi.org/10.1016/j.aos.2006.08.001
- Alchian, A.A. and Demsetz, H. (1972), "Production, information costs, and economic organization", *The American Economic Review*, Vol. 62 No. 5, pp. 777–795.
- Aleksovska, M., Schillemans, T. and Grimmelikhuijsen, S. (2019), "Lessons from five decades of experimental and behavioral research on accountability: a systematic literature review", *Journal of Behavioral Public Administration*, Vol. 2 No. 2. https://doi.org/10.30636/jbpa.22.66
- Allbon, C. (2012), "Down the rabbit hole"—"curiouser and curiouser": using autoethnography as a mode of writing to re-call, re-tell and re-veal bodily embodiment as self-reflexive inquiry", *Journal of Organizational Ethnography*, Vol. 1 No. 1, pp. 62–71. https://doi.org/10.1108/20466741211220660
- Amit, V. (2003), "The university as panopticon: moral claims and attacks on academic freedom", Strathern, M. (Ed.), *Audit Cultures: Anthropological Studies in Accountability, Ethics and the Academy*, Routledge, London, pp. 227–247.
- Anderson, L. (2006), "Analytic autoethnography", *Journal of Contemporary Ethnography*, Vol. 35 No. 4, pp. 373–395. https://doi.org/10.1177/0891241605280449
- Arbin, K., Frostenson, M., Helin, S. and Borglund, T. (2021), "Explaining workers' resistance against a health and safety programme: an understanding based on hierarchical and social accountability", *Safety Science*, Vol. 136, art. 105131. https://doi.org/10.1016/j.ssci.2020.105131

- Armstrong, P. (2008), "Calling out for more: comment on the future of interpretive accounting research", *Critical Perspectives on Accounting*, Vol. 19 No. 6, pp. 867–879. https://doi.org/10.1016/j.cpa.2007.02.010
- Arnold, E., Simmonds, P., Farla, K., Kolarz, P., Mahieu, B. and Nielsen, K. (2018), *Review of the Research Excellence Framework: Evidence Report*, Technopolis Group, Brighton, UK.
- Atkins, J. and Maroun, W. (2018), "Integrated extinction accounting and accountability: building an ark", *Accounting, Auditing & Accountability Journal*, Vol. 31 No. 3, pp. 750–786. https://doi.org/10.1108/AAAJ-06-2017-2957
- Atkinson, P. (1997), "Narrative turn or blind alley?", *Qualitative Health Research*, Vol. 7 No. 3, pp. 325–344. https://doi.org/10.1177/104973239700700302
- Barkas, L.A., Scott, J.M., Poppitt, N.J. and Smith, P.J. (2019), "Tinker, tailor, policy-maker: can the UK government's teaching excellence framework deliver its objectives?", *Journal of Further and Higher Education*, Vol. 43 No. 6, pp. 801–813. https://doi.org/10.1080/0309877X.2017.1408789
- BBC (2021), "University strikes: students react as staff walk out", *BBC News*, December 1. https://www.bbc.co.uk/news/education-59477569
- Behar, R. (1997), *The Vulnerable Observer: Anthropology That Breaks Your Heart*, 1st edition, Beacon Press, Boston, MA.
- Boje, D. and Tyler, J.A. (2009), "Story and narrative noticing: workaholism autoethnographies", *Journal of Business Ethics*, Vol. 84, pp. 173–194. https://doi.org/10.1007/s10551-008-9702-7
- Brown, J., Kirk-Wade, E. and Barber, S. (2021), "Coronavirus: a history of 'lockdown laws' in England", *House of Commons Library*, December 22.
- Butler, J. (2005), *Giving an Account of Oneself*, annotated edition, Fordham University Press, New York, NY.
- Cohen, A.R. (1976), "Beyond simulation: treating the classroom as an organization", *The Teaching of Organizational Behavior*, Vol. 2 No. 1, pp. 13–19. https://doi.org/10.1177/105256297600200103
- Cooper, C. and Lapsley, I. (2021), "Hillsborough: the fight for accountability", *Critical Perspectives on Accounting*, Vol. 79, art. 102077. https://doi.org/10.1016/j.cpa.2019.02.004
- Cordery, C., Belal, A.R. and Thomson, I. (2019), "NGO accounting and accountability: past, present and future", *Accounting Forum*, Vol. 43 No. 1, pp. 1–15. https://doi.org/10.1080/01559982.2019.1593577
- Dalla Via, N., Perego, P. and Van Rinsum, M. (2019), "How accountability type influences information search processes and decision quality", *Accounting, Organizations and Society*, Vol. 75, pp. 79–91. https://doi.org/10.1016/j.aos.2018.10.001
- Deem, R. and Baird, J.-A. (2020), "The English teaching excellence (and student outcomes) framework: intelligent accountability in higher education?", *Journal of Educational Change*, Vol. 21, pp. 215–243. https://doi.org/10.1007/s10833-019-09356-0
- Doloriert, C. and Sambrook, S. (2009), "Ethical confessions of the 'I' of autoethnography: the student's dilemma", *Qualitative Research in Organizations and Management*, Vol. 4 No. 1, pp. 27–45. https://doi.org/10.1108/17465640910951435
- Edelman, M. (1996), "Devil, not-quite-white, rootless cosmopolitan: Tsuris in Latin America, the Bronx, and the USSR", Ellis, C. and Bochner, A. P. (Eds.), *Composing Ethnography: Alternative Forms of Qualitative Writing*, AltaMira, Lanham, MD, pp. 267–300.
- Ellis, C. (2009), "Telling tales on neighbors: ethics in two voices", *International Review of Qualitative Research*, Vol. 2 No. 1, pp. 3–27. https://doi.org/10.1525/irqr.2009.2.1.3
- Ellis, C. and Bochner, A. (2000), "Autoethnography, personal narrative, reflexivity: researcher as subject", Denzin, N.K. and Lincoln, Y.S (Eds.), *Handbook of Qualitative Research*, 2nd edition, Sage Publications, Thousand Oaks, CA, pp. 733–768.

- Etherington, K. (2004), *Becoming a Reflexive Researcher: Using Our Selves in Research*, Jessica Kingsley, London.
- Fillitz, T. (2003), "Academia: same pressures, same conditions of work?", Strathern, M. (Ed.), *Audit Cultures: Anthropological Studies in Accountability, Ethics and the Academy*, Routledge, London, pp. 236–255.
- Foucault, M. (1979), *Discipline and Punish: The Birth of the Prison*, Vintage Books, New York, NY.
- Fry, R.E. (1995), "Accountability in organizational life: problem or opportunity for nonprofits?", *Nonprofit Management and Leadership*, Vol. 6 No. 2, pp. 181–195. https://doi.org/10.1002/nml.4130060207
- Gefou-Madianou, D. (2003), "Disciples, discipline and reflection: anthropological encounters and trajectories", Strathern, M. (Ed.), *Audit Cultures: Anthropological Studies in Accountability, Ethics and the Academy*, Routledge, London, pp. 248–267.
- Griffith, A.S. and Altinay, Z. (2020), "A framework to assess higher education faculty workload in US universities", *Innovations in Education and Teaching International*, Vol. 57 No. 6, pp. 691–700. https://doi.org/10.1080/14703297.2020.1786432
- Grisard, C., Graham, C. and Annisette, M. (2017), "Negotiations of accounting figures and the individualizing effects of hierarchical accountability", paper presented at the European Accounting Association (EAA) Annual Conference, Valencia, May 10–12.
- Habermas, J. (1971), *Toward a Rational Society: Student Protest, Science, and Politics*, 1st edition, Beacon Press, Boston, MA.
- Habermas, J. (1985), *The Theory of Communicative Action, Volume 2: Lifeworld and System:* A Critique of Functionalist Reason, Beacon Press, Boston, MA.
- Hall, D. and Buzwell, S. (2013), "The problem of free-riding in group projects: looking beyond social loafing as reason for non-contribution", *Active Learning in Higher Education*, Vol. 14 No. 1, pp. 37–49. https://doi.org/10.1177/1469787412467123
- Haynes, K. (2011), "Tensions in (re)presenting the self in reflexive autoethnographical research", *Qualitative Research in Organizations and Management*, Vol. 6 No. 2, pp. 134–149. https://doi.org/10.1108/17465641111159125
- Haynes, K. (2013), "Sexuality and sexual symbolism as processes of gendered identity formation: an autoethnography of an accounting firm", *Accounting, Auditing & Accountability Journal*, Vol. 26 No. 3, pp. 374–398. https://doi.org/10.1108/09513571311311865
- Holman-Jones, S. (2005), "Autoethnography: making the personal political", Denzin, N.K. and Lincoln, Y.S (Eds.), *Handbook of Qualitative Research*, 2nd edition, Sage Publications, Thousand Oaks, CA, pp. 763–791.
- Holt, N.L. (2003), "Representation, legitimation, and autoethnography: an autoethnographic writing story", *International Journal of Qualitative Methods*, Vol. 2 No. 1, pp. 18–28. https://doi.org/10.1177/160940690300200102
- Hood, C. (2007), "What happens when transparency meets blame-avoidance?", *Public Management Review*, Vol. 9 No. 2, pp. 191–210. https://doi.org/10.1080/14719030701340275
- Hoskin, K. (1996), "The 'awful idea of accountability': inscribing people into the measurement of objects", Mouritsen, J. and Munro, R. (Eds.), *Accountability: Power, Ethos and the Technologies of Managing*, Thomson, London, pp. 265–282.
- Humphreys, M. (2005), "Getting personal: reflexivity and autoethnographic vignettes", *Qualitative Inquiry*, Vol. 11 No. 6, pp. 840–860. https://doi.org/10.1177/1077800404269425

- Jacobs, K. (2016), "Theorising interdisciplinary public sector accounting research", *Financial Accountability & Management*, Vol. 32 No. 4, pp. 469–488. https://doi.org/10.1111/faam.12093
- Kou, C.-Y. and Stewart, V. (2018), "Group accountability: a review and extension of existing research", *Small Group Research*, Vol. 49 No. 1, pp. 34–61. https://doi.org/10.1177/1046496417712438
- Lai, A., Leoni, G. and Stacchezzini, R. (2014), "The socializing effects of accounting in flood recovery", *Critical Perspectives on Accounting*, Vol. 25 No. 7, pp. 579–603. https://doi.org/10.1016/j.cpa.2014.04.002
- Lai, A., Melloni, G. and Stacchezzini, R. (2018), "Integrated reporting and narrative accountability: the role of preparers", *Accounting, Auditing & Accountability Journal*, Vol. 31 No. 5, pp. 1381–1405. https://doi.org/10.1108/AAAJ-08-2016-2674
- Lane, D.R. (2012), "Peer feedback processes and individual accountability in team-based learning", Sweet, M. and Michaelson, L.K. (Eds.), *Team-Based Learning in the Social Sciences and Humanities*, Stylus, Sterling, VA, pp. 51–62.
- Leoni, G., Lai, A., Stacchezzini, R., Steccolini, I., Brammer, S., Linnenluecke, M. and Demirag, I. (2021), "Accounting, management and accountability in times of crisis: lessons from the COVID-19 pandemic", *Accounting, Auditing & Accountability Journal*, Vol. 34 No. 6, pp. 1305–1319. https://doi.org/10.1108/AAAJ-05-2021-5279
- Lillejord, S. (2020), "From 'unintelligent' to intelligent accountability", *Journal of Educational Change*, Vol. 21, pp. 1–18. https://doi.org/10.1007/s10833-020-09379-y
- Lindkvist, L. and Llewellyn, S. (2003), "Accountability, responsibility and organization", *Scandinavian Journal of Management*, Vol. 19 No. 2, pp. 251–273. https://doi.org/10.1016/S0956-5221(02)00027-1
- Llewellyn, S. (1994), "Managing the boundary: how accounting is implicated in maintaining the organization", *Accounting, Auditing & Accountability Journal*, Vol. 7 No. 4, pp. 4–23. https://doi.org/10.1108/09513579410069821
- Marginson, D. and Ogden, S. (2005), "Coping with ambiguity through the budget: the positive effects of budgetary targets on managers' budgeting behaviours", *Accounting, Organizations and Society*, Vol. 30 No. 5, pp. 435–456. https://doi.org/10.1016/j.aos.2004.05.004
- McKie, A. (2020), "One in three union members joined strikes, say employers", *Times Higher Education*, January 7. https://www.timeshighereducation.com/news/one-three-union-members-joined-strikes-say-employers
- Messner, M. (2009), "The limits of accountability", *Accounting, Organizations and Society*, Vol. 34 No. 8, pp. 918–938. https://doi.org/10.1016/j.aos.2009.07.003
- Mezoff, R.M., Cohen, A.R. and Bradford, D.L. (1979), "A dialogue on 'treating the classroom as an organization", *Exchange: The Organizational Behavior Teaching Journal*, Vol. 4 No. 1, pp. 25–36. https://doi.org/10.1177/105256297900400106
- Mulgan, R. (2000), "'Accountability': an ever-expanding concept?", *Public Administration*, Vol. 78 No. 3, pp. 555–573. https://doi.org/10.1111/1467-9299.00218
- Munro, R. (1996), "Alignment and identity work: the study of accounts and accountability", Munro, R. and Mouritsen, J. (Eds.), *Accountability: Power, Ethos and the Technologies of Managing*, Thomson, London, pp. 1–19.
- Munro, R.J.B. and Hatherly, D.J. (1993), "Accountability and the new commercial agenda", *Critical Perspectives on Accounting*, Vol. 4 No. 4, pp. 369–395. https://doi.org/10.1006/cpac.1993.1020
- O'Neill, O. (2002), *A Question of Trust: The BBC Reith Lectures 2002*, 2nd edition, Cambridge University Press, Cambridge.

- O'Neill, O. (2013), "Intelligent accountability in education", *Oxford Review of Education*, Vol. 39 No. 1, pp. 4–16. https://doi.org/10.1080/03054985.2013.764761
- O'Regan, B., King, R. and Smith, D. (2021), "Combining accountability forms: transparency and 'intelligent' accountability in a public service organization", *Accounting, Auditing & Accountability Journal*, Vol. 35 No. 5, pp. 1287–1315. https://doi.org/10.1108/AAAJ-03-2020-4473
- Office for Students (n.d.), "National Student Survey NSS", *Office for Students* [website]. URL https://www.officeforstudents.org.uk/advice-and-guidance/student-information-and-data/national-student-survey-nss/
- Office for Students (2020), "The National Student Survey: consistency, controversy and change", Office for Students Insight Brief 6, February. https://www.officeforstudents.org.uk/publications/the-national-student-survey-consistency-controversy-and-change/
- Office for Students (2022), "Maintaining the credibility of degrees", Office for Students Insight Brief 14, July. https://www.officeforstudents.org.uk/media/a712215d-94e8-4e98-8614-baee4d5aa3dc/insight-brief-14.pdf
- Oliveira, G., Grenha Teixeira, J., Torres, A. and Morais, C. (2021), "An exploratory study on the emergency remote education experience of higher education students and teachers during the COVID-19 pandemic", *British Journal of Educational Technology*, Vol. 52 No. 4, pp. 1357–1376. https://doi.org/10.1111/bjet.13112
- Pendse, S. (1984), "E pluribus unum: making a classroom an organization", *Organizational Behavior Teaching Review*, Vol. 9 No. 4, pp. 41–51. https://doi.org/10.1177/105256298400900407
- Pierce, S. (2021), *Independent Review of the Teaching Excellence and Student Outcomes Framework (TEF): Report to the Secretary of State for Education*, UK Government, London. https://www.gov.uk/government/publications/independent-review-of-tef-report
- Pitman, T. (2016), "The evolution of the student as a customer in Australian higher education: a policy perspective", *Australian Educational Researcher*, Vol. 43, pp. 345–359. https://doi.org/10.1007/s13384-016-0204-9
- Power, M. (2018), "Accounting, boundary-making, and organizational permeability", Ringel, L., Hiller, P. and Zietsma, C. (Eds.), *Toward Permeable Boundaries of Organizations?*, Emerald Publishing, Bingley, pp. 31–53. https://doi.org/10.1108/S0733-558X20180000057002
- Radice, H. (2013), "How we got here: UK higher education under neoliberalism", *ACME: An International Journal for Critical Geographies*, Vol. 12 No. 2, pp. 407–418.
- Rahaman, A., Neu, D. and Everett, J. (2010), "Accounting for social-purpose alliances: confronting the HIV/AIDS pandemic in Africa", *Contemporary Accounting Research*, Vol. 27 No. 4, pp. 1093–1129. https://doi.org/10.1111/j.1911-3846.2010.01040.x
- Ricoeur, P. (1995), Oneself as Another, University of Chicago Press, Chicago, IL.
- Rimoldi, E. (2003), "Generic genius—how does it all add up?", Strathern, M. (Ed.), *Audit Cultures: Anthropological Studies in Accountability, Ethics and the Academy*, Routledge, London, pp. 102–117.
- Roberts, J. (1991), "The possibilities of accountability", *Accounting, Organizations and Society*, Vol. 16 No. 4, pp. 355–368. https://doi.org/10.1016/0361-3682(91)90027-C
- Roberts, J. (1996), "From discipline to dialogue: individualizing and socializing forms of accountability", Munro, R. and Mouritsen, J. (Eds.), *Accountability: Power, Ethos and the Technologies of Managing*, Cengage, Andover, pp. 40–61.
- Roberts, J. (2003), "The manufacture of corporate social responsibility: constructing corporate sensibility", *Organization*, Vol. 10 No. 2, pp. 249–265. https://doi.org/10.1177/1350508403010002004

- Roberts, J. (2009), "No one is perfect: the limits of transparency and an ethic for 'intelligent' accountability", *Accounting, Organizations and Society*, Vol. 34 No. 8, pp. 957–970. https://doi.org/10.1016/j.aos.2009.04.005
- Roberts, J. (2018), "Accountability", Roslender, R. (Ed.), *The Routledge Companion to Critical Accounting*, Routledge, Oxford, pp. 225–239.
- Roberts, J. and Scapens, R. (1985), "Accounting systems and systems of accountability: understanding accounting practices in their organisational contexts", *Accounting, Organizations and Society*, Vol. 10 No. 4, pp. 443–456. https://doi.org/10.1016/0361-3682(85)90005-4
- Robson, K. (1992), "Accounting numbers as 'inscription': action at a distance and the development of accounting", *Accounting, Organizations and Society*, Vol. 17 No. 7, pp. 685–708. https://doi.org/10.1016/0361-3682(92)90019-O
- Sadler, E. and Wessels, J.S. (2019), "Transformation of the accounting profession: an autoethnographical reflection of identity and intersectionality", *Meditari Accountancy Research*, Vol. 27 No. 3, pp. 448–471. https://doi.org/10.1108/MEDAR-05-2018-0339
- Saldivar, K.M. (2015), "Team-based learning: a model for democratic and culturally competent 21st century public administrators", *Journal of Public Affairs Education*, Vol. 21 No. 2, 143–164. https://doi.org/10.1080/15236803.2015.12001825
- Scobie, M.R., Milne, M.J. and Love, T.R. (2020), "Dissensus and democratic accountability in a case of conflict", *Accounting, Auditing & Accountability Journal*, Vol. 33 No. 5, pp. 939–964. https://doi.org/10.1108/AAAJ-11-2016-2780
- Shore, C. and Wright, S. (2003), "Coercive accountability: the rise of audit culture in higher education", Strathern, M. (Ed.), *Audit Cultures: Anthropological Studies in Accountability, Ethics and the Academy*, Routledge, London, pp. 69–101.
- Sinclair, A. (1995), "The chameleon of accountability: forms and discourses", *Accounting, Organizations and Society*, Vol. 20 Nos. 2–3, pp. 219–237. https://doi.org/10.1016/0361-3682(93)E0003-Y
- Smith, D.E. and Fleisher, S. (2011), "The implications of grade inflation: faculty integrity versus the pressure to succeed", *Journal of Research in Innovative Teaching*, Vol. 4, pp. 32–38.
- Stein, R.E., Colyer, C.J. and Manning, J. (2016), "Student accountability in team-based learning classes", *Teaching Sociology*, Vol. 44 No. 1, pp. 28–38. https://doi.org/10.1177/0092055X15603429
- Strathern, M. (2003), "New accountabilities: anthropological studies in audit, ethics and the academy", Strathern, M. (Ed.), *Audit Cultures: Anthropological Studies in Accountability, Ethics and the Academy*, Routledge, London, pp. 13–30.
- Streeting, W. and Wise, G. (2009), *Rethinking the Values of Higher Education: Consumption, Partnership, Community?*, QAA, London. http://dera.ioe.ac.uk/434/2/Rethinking.pdf
- Swansea University (2022), "Academic career pathways: Swansea University", Swansea University [website]. https://www.swansea.ac.uk/jobs-at-swansea/performance-enabling/academic-career-pathways/
- Times Higher Education (2017), "Teaching excellence framework (TEF) results 2017", *THE*, June 22. https://www.timeshighereducation.com/news/teaching-excellence-framework-tefresults-2017
- Topping, K. (1998), "Peer assessment between students in colleges and universities", *Review of Educational Research*, Vol. 68 No. 3, pp. 249–276. https://doi.org/10.3102/00346543068003249
- Turner, B. (1990), "The rise of organizational symbolism", Hassard, J. and Pym, D. (Eds.), *The Theory and Philosophy of Organizations: Critical Issues and New Perspectives*, Routledge, London, pp. 83–96.

- UCU (2010), Student Participation in Quality Assurance in the FE and HE Sectors, University and College Union, London.
- UCU (2016), "UCU and the National Union of Students statement on the National Student Survey boycott", *UCU*, December 14. https://www.ucu.org.uk/article/8571/UCU-and-the-National-Union-of-Students-statement-on-the-National-Student-Survey-boycott
- UCU (2022a), "Biggest ever university strikes set to hit UK campuses over pay, conditions & pensions", *UCU*, November 8. https://www.ucu.org.uk/article/12609/Biggest-ever-university-strikes-set-to-hit-UK-campuses-over-pay-conditions-pensions
- UCU (2022b), Workload Survey 2021 Data Report, University and College Union, London.
- Vollmer, H. (2007), "How to do more with numbers: elementary stakes, framing, keying, and the three-dimensional character of numerical signs", *Accounting, Organizations and Society*, Vol. 32 No. 6, pp. 577–600. https://doi.org/10.1016/j.aos.2006.10.001
- Webster, K. (2012), "The evolving role of libraries in the scholarly ecosystem", Campbell, R., Pentz, E. and Borthwick, I. (Eds.), *Academic and Professional Publishing*, Chandos Publishing, Cambridge, pp. 315–336.
- Wilkin, C.L. (2022), "Developing critical reflection: an integrated approach", *The British Accounting Review*, Vol. 54 No. 3, art. 101043. https://doi.org/10.1016/j.bar.2021.101043
- Yu, A. (2021), "Accountability as mourning: accounting for death in the time of COVID-19", *Accounting, Organizations and Society*, Vol. 90, art. 101198. https://doi.org/10.1016/j.aos.2020.101198

 Table 1. Peer feedback template

•	Peer Evaluation Score
Equal Share + 50%	150%
Equal Share + 40%	140%
Equal Share + 30%	130%
Equal Share + 20%	120%
Equal Share + 10%	110%
Equal Share	100%
Equal Share - 10%	90%
Equal Share - 20%	80%
Equal Share - 30%	70%
Equal Share - 40%	60%
Equal Share - 50%	50%
0	0%